Author's response to reviews

Title: In Vitro and In Vivo antifungal activities of selected Cameroonian dietary spices

Authors:

Jean Paul Dzoyem (jpdzoyem@yahoo.fr)
Roland T. Tchuenguem (tchuenguemroland@yahoo.com)
Jules R. Kuiate (irkuiate@yahoo.com)
Gerald N. Teke (ngobob@yahoo.com)
Frederick Agem Kechia (kechiafred@yahoo.fr)
Victor Kuete (kuetevictor@yahoo.fr)

Version: 6  Date: 5 February 2014

Author's response to reviews:

Answer to Editorial comments:

Confirm whether any guidelines of this type were utilised during the course of your study. If so, then a statement to this effect should be included in your Methods section.

Answer: Thank you Editor, as requested the guidelines followed during our study has been addressed. A detailed statement describing the guidelines has been included in the method section.